Author's response to reviews

Title: Development and promotion of a national website to improve dissemination of information related to the prevention of mother-to-child HIV transmission (PMTCT) in Tanzania

Authors:

Gudila Stephan (gudilastephan@yahoo.com)
Mary Jo Hoyt (hoyt@sn.rutgers.edu)
Deborah Storm (stormds@sn.rutgers.edu)
Sylvia Shirima (shrimsy@fxbtanzania.org)
Charles Matiko (Matiko@fxbtanzania.org)
Emmanuel Matechi (matechi@fxbtanzania.org)

Version: 2
Date: 25 August 2014

Author's response to reviews:

This manuscript was initially submitted to the journal, Human Resources for Health and received the following response. “The Editors have assessed your manuscript...[and] although interesting your manuscript is not exactly within the scope of the journal. The Editors suggested that you might consider submitting your manuscript to journal with a broader focus on public health. The Editors believe BMC Public Health would be appropriate for your research.” We appreciated this guidance and are very pleased to submit our manuscript for consideration by BMC Public Health. We have reviewed and made minor edits in the text and its arrangement, but have not made substantive content changes in the manuscript.